\begin{table}%t2
%\centering
\par
\caption{Table of adopted acronyms. For each adopted acronym we give
an example of a name and the reference where the acronym was created.
When an acronym comes from several references (e.g., KUG), we give
the latest reference that lists the former ones. In the bibliography, 
the full reference is given with "Cat'' followed by the acronym 
(e.g., Cat.: GIN).}
\begin{tabular}{ll}
\hline\hline
Acronym (with example)        &  Reference                                 \\
\hline
   ABELL 0592:$[D80]$ 012     &  Dressler A. (\cite{Dressler80})                         \\
   ARK 7                      &  Arakelian M.A. (\cite{Arakelian75})                      \\
   ARP 223                    &  Arp, H. (\cite{Arp66})                             \\
   CGMW 1-0056                &  Roman A.T. et~al. (\cite{Roman00})                   \\
   $[DCL86]$ 571              &  Dickens R.J. et~al. (\cite{Dickens86})                 \\
   DDO  207                   &  van den Bergh S. (\cite{Bergh60})                    \\
   DUKST 474-11               &  Ratcliffe A. et~al. (\cite{Ratcliffe98})                 \\
   ESO  123-012               &  Lauberts A. (\cite{Lauberts82})                         \\
   ...                        &  ...                                       \\
   ...                        &  ...                                       \\
   ...                        &  ...                                       \\
\hline
\end{tabular}
\label{acrosln}
\end{table}